Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: It is an interesting case and has clinical guiding significance. It would be better if you could make some modifications

1. In the section of History of past illness, the time of history of hypertension should be clear in the sentence of “She also had hypertension.” The correlation between hypertension and forearm compartment syndrome could also be discussed.

**Author response**
I added the onset of hypertension. Since we found the inaccurate description about the age of surgery, I corrected this as well.

**Author action**
Page 7, line 9: When the patient was 70 years old, she underwent surgery for left breast cancer. She also had hypertension. → When the patient was 73 years old, she underwent surgery for left breast cancer. She also had hypertension since she was 70 years old.

2. In the section Physical examination, it will be more consistent if “The volar compartment pressure” could be changed to “The pressure of volar compartment”.

**Author response**
I changed the phrase.

**Author action**
Page 7, line 14: The volar compartment pressure → The pressure of volar compartment

3. Activated partial thromboplastin time (aPTT) and APTT should be consistent in the context.

**Author response**
I changed the following word.

**Author action**
Page 8, line 11: APTT → aPTT

4. If available, it will be better to show the pictures of bilateral forearms, and lower abdomen, and right proximal thigh at the end of treatment.

**Author response**
Because I did not take the pictures at the end of treatment, it is difficult to include them.